\documentclass{amsart}  % {{{
\usepackage[breaklinks, final]{hyperref}

% misc
\setlength{\parindent}{0pt} % no paragraph indents
\setlength{\parskip}{5pt}   % double space
\pagestyle{empty}           % no page numbers

% document specific stuff

%============================================================================}}}

\begin{document}
\title{EECS 210: Discrete structures \\
  discussion session, sections 1400, 1800, 1900 \\
  Spring 2018}
\maketitle
\thispagestyle{empty}

\textbf{Teaching Assistant:} Casey Sader

\textbf{Office:} Eaton 2041

\textbf{Office Hours:} Mon.\ 2:00-3:00 PM, Wed.\ 3:00-4:00 PM, and by
appointment (Times could change)

\textbf{E-Mail:} \href{mailto:c455s614@ku.edu} {\nolinkurl{c455s614@ku.edu}}

\textbf{Discussion Times and Venues:} \\
Section 1400: Monday 4:30pm -- 6:20pm, Learned 2111 \\
Section 1800: Tuesday 11:00am -- 12:50pm, Learned 2115 \\
Section 1900: Tuesday 4:30pm -- 6:20pm, Learned 1131

\textbf{Honor Code:} All work submitted for credit must be the student's
own. If you have worked with another student on a homework problem, you
should indicate it in your solution. Such cooperation is permitted, but your
work must be your own. Quizzes and exams should be completed without
unauthorized aid.

\textbf{Accommodation Procedure:} The Academic Achievement and Access Center
(AAAC) coordinates academic accommodations and services for all eligible KU
students with disabilities. If you have a disability for which you wish to
request accommodations and have not contacted the AAAC, please do so as soon
as possible. Information about their services can be found at
\url{http://www.access.ku.edu}. Please contact me privately in regard to
your needs in this course.

\textbf{Religious Observances:} Any student in this course who plans to
observe a religious holiday which conflicts in any way with the course
schedule or requirements should contact me as soon as possible to discuss
alternative accommodations.
\end{document}
